Great Basin Incident Commander Conference Call Agenda 2016

Date: Saturday, August 27, 2016

| Note taker: Julie Campbell

GB MAC/GACC Coordinator: Derrek Hartman
Agenda/Incident Commander Discussion Points: Summarize the following in 4 min or less:

. Fire Size / % contained / # of people / Cost to date

. Significant changes since the last 209

. BRIEF fire synopsis /Critical Incident Information (use Directional vs DIVS’s)
. Political Issues / Community Meetings

o Closures / Evacuations / Public Safety Issues

. Major Safety Concerns / Injuries / Accidents

. Critical Resource Needs

. Resources Available to Reassign/Share/Demob

Predictive Services:
Weather update:
Shelby

Breezy winds today in Eastern ID and Western WY. High Pressure in place so
increase of temperature and winds, however Pacific NW low pressure
coming so will be an increase of SW winds next week. Cooler Temperatures
maybe by next weekend-stay tuned

GBCC Update: e GBCCatPL4,NICCPL4
Ray Crowe e 10 new fires for 76 acres
e Large fire growth 2,247 acres
MAC / GBCC Report from the ASAT. Today Idaho Falls and Pocatello. Tomorrow Jackson.

Coordinator
Derrek Hartman

No concerns reported.
High demand for crews but supply low-

1
Fire: Henry’s Creek
IC: DeMasters

52,511 /55% / 443 / 2,900,000.00 (increase of 1 ac)

The S and SW side of fire looking pretty secure, holding and mopping as
needed. Since there is significant PVT lands effected, will be working on a
rehab plan today. Activity on the N/NW-working indirect with dozers/hand
line has been successful and threat has diminished. East side still seeing
some moderate fire behavior so working hand line there. Currently
county/FS closures however no fire on FS lands. Threats continue to be
trout, sage grouse, structures, sheep/cows and the Bonneville Powerlines.
Concerns with hunters and the holiday. Did lose 1 structure last Tuesday
that was not confirmed until yesterday.

Critical needs; 3 RADO’s

Resources to reassign T1 CWN S61 Helicopter and 2 T2IA crews tomorrow.
FMAG approved for this fire-working on cost share

2

Fire: Pioneer
IC: Lund

IC (t) Adell

104,475 / 58% / 1547 / 71,000,000.00

Good day yesterday. Securing, mopping up and backhaul on west side. S/SE
side repair to dozer line continues-huge back haul work being worked on.
North side continues to be active and with weather continues to push up
fingers. Held to the west side of Clear Creek drainage. Still concerns with
summer homes. Intent is still to keep it out of Long Creek drainage.
Community meeting with Garden Valley last night. Overall went well.




3
Berry
IC: Roide

12,378 / 20% /351/$1,600,000.00

Working on Point Protection and Perimeter control, particularly focusing on
west side of river to reduce threat to Flagg Ranch . Major Hwy connecting
Grant Teton and south entrance of Yellowstone National Park remains
closed. Flagg Ranch and 2 campgrounds remain evacuated. Structure
protection for those entities. Working on Long Term Plans. Otherwise fire
will be point protection. Political issues are Hwy 89 closure.

Injury/lliness-1 discomfort for 2 days, taken in but released, light duty today
Red Flag Conditions today.

4
Tie
IC: Bird

1,260 / 25% / 250 / 1,035,000.

Moderate fire behavior with spotting. Mopping up North Side. Concentrated
on E side with direct hand line with support of scoopers and helicopters.
Concerns are predicted winds, a major powerline, hunting season
approaching and Hwy 31 (scenic byway from Swan Valley to Victor, ID), L1
evacuation for 20 residences located just SW of Victor, ID. Some trail
closures on FS lands. Meeting last night in Victor, ID went very well.
Accidents; one forklift making contact with a rental vehicle.

Report of a repeater box being transported via helicopter and the antenna
cable within the box shorted out and actually burnt the paper inside the
box. Repeater box still in working condition. Safenet and Safecom being
done.

Critical needs; 2 T1 and 2 T2 IA crews, 3 engines (no typing) and 1 camp
crew

5

Fire: Peterson Hollow
IC: Darren Turner
John Platt (t)

1,200/ 20% / 290 / $1,900,000.00

Moderate fire behavior with some single tree torching. Meeting objectives
to the west. South end going direct and indirect. North end doing some
good work as well. Supported UWC with IA yesterday. Safety concerns
continue to be snags, sink holes, smoke impacts along Hwy. and fire
behavior. Critical needs; 3 T2IA crews, 2 T1 helicopters, AT and Scoopers

6 3,418 / 50% / 60/ $1,070,000.00

Fire: Briggs Working on contingency line and will fire out if conditions warrant.
IC: Osler Plan to turn over fire to a T4 on Tuesday at 1000.

7 3 Fires are being managed-

Fires: South Fork Zone
Big Elk, Carrot, Black
Fires

IC: Deb Flowers

Big Elk 68 acres, inactive, no concerns

Carrot 218 acres NE of Driggs, ID in the Jedidiah Smith Wilderness,
monitoring.

Black Fire 300+ acres north Swan Valley, ID. Scouting for possible
contingency line work. Area and trail closures on all 3 fires. Fires monitoring
via web cam. Total personnel 50

Critical needs; T2 IA crews so be successful to meet objectives.

8
Fire: Box Canyon
IC: Hillman

3,947 / 20/ 40/ 3,000,000.00

Fire was T3 turned to a T4 but reached management actions points so T3
was put back into place yesterday. Evacuated Ledge Fork Campground.
Working direct and indirect. Spot in Hell’s Kitchen so working that.
Critical needs; T1 helicopter




Date/Time of Next Call:

8/28 @ 0800 MT /0700 PT




